Author's response to reviews

Title: Distal end side-to-side anastomoses of sequential vein graft to small target coronary arteries improve graft patency

Authors:

Haitao Li (liht987@163.com)
Baodong Xie (kangkai@sina.com)
Chengxiong Gu (anzhengu@sina.com)
Mingxin Gao (star880312@126.com)
Fan Zhang (zf85715@163.com)
Jiayang Wang (athlandwang@163.com)
Longsheng Dai (dlsdailongsheng@126.com)
Yang Yu (heartyuyang@hotmail.com)

Version: 2
Date: 28 January 2014

Author's response to reviews:

Dear Erica,

To address the issues you have brought up, we have added author's contribution section and email information for all the authors. The figure legend has been moved after the tables.

Thank you

Sincerely,

Yang Yu